Author’s response to reviews

Title: Pregnancy as a Risk Factor for Severe Influenza Infection: An Individual Participant Data Meta-Analysis

Authors:
Dominik Mertz (mertzd@mcmaster.ca)
Calvin Lo (calvinlo66@gmail.com)
Lyubov Lytvyn (lytvyn.lyubov@gmail.com)
Justin Ortiz (JOrtiz@som.umaryland.edu)
Mark Loeb (loebm@mcmaster.ca)

Version: 2 Date: 17 Jul 2019

Author’s response to reviews:
All editorial requests addressed, i.e. changing the headings including addition of a subheading "conclusions", added a list of abbreviations before the declarations section, and added a "Consent for publication" section.